Author’s response to reviews

Title: In-screw polymethylmethacrylate-augmented sacroiliac screw for the treatment of fragility fractures of the pelvis: a prospective, observational study with 1-year follow-up

Authors:

Andreas Höch (andreas.hoech@medizin.uni-leipzig.de)
Philipp Pieroh (philipp.pieroh@medizin.uni-leipzig.de)
Ralf Henkelmann (ralf.henkelmann@medizin.uni-leipzig.de)
Christoph Josten (Christoph.Josten@medizin.uni-leipzig.de)
Jörg Böhme (kerstin.stadler@sanktgeorg.de)

Version: 2 Date: 09 Oct 2017

Author’s response to reviews:

BSUR-D-17-00226

In-screw polymethylmethacrylate-augmented sacroiliac screw for fragility fractures of the pelvis: a prospective, observational study with 1-year follow-up

Andreas Höch; Philipp Pieroh; Ralf Henkelmann; Christoph Josten; Jörg Böhme

BMC Surgery

Dear Editor,

First of all we thank you for the opportunity to revise our manuscript and therefore we thank you for your valuable recommendations to improve the manuscript.

In the follow you find a point-to-point response to these comments. Changes in the manuscript made are highlighted.
We hope your concerns were addressed adequately and the manuscript is suitable for publication in your esteemed journal.

Sincerely yours,

Andreas Höch and Philipp Pieroh on behalf of all authors